Author's response to reviews

Title: Appropriate interventions for the prevention and management of self-harm: a qualitative exploration of service-users views

Authors:

Megan E Hume (megan.hume@ed.ac.uk)
Stephen Platt (steve.platt@ed.ac.uk)

Version: 3 Date: 19 January 2007

Author's response to reviews:

Dear Sir/Madam,

Following your email regarding core revisions, I am pleased to submit a final report to you. The following changes have been made in light of your formatting requests, and 1 change in response to an reviewer's comment.

1. 'The country of affiliation has been added'. pg 1.
2. The authors declare no competing interests. pg 23
3. 'Both authors read and approved the final manuscript'. pg 23
4. In light of Allan House's comment about the Barnes Unit, we have removed this reference and utilised his helpful suggestion of an alternative reference (ref. 16).
"The findings support the appropriateness of the move towards the provision of this type of integrated support for people who self-harm". Pg. 19.

Yours Faithfully,

Megan Hume
Research Assistant
Research Unit in Health, Behaviour and Change
University of Edinburgh
Medical School
Teviot Place
Edinburgh, EH8 9AG
Megan.hume@ed.ac.uk
0131 650 4619